**Author’s response to reviews**

**Title:** BURNOUT IN MEDICAL STUDENTS: A SYSTEMATIC REVIEW OF EXPERIENCES IN CHINESE MEDICAL SCHOOLS

**Authors:**

Wang Chunming (trillionwang@gmail.com)

Reema Harrison (reema.harrison@unsw.edu.au)

Raina MacIntyre (r.macintyre@unsw.edu.au)

Joanne Travaglia (joanne.travaglia@uts.edu.au)

Chinthaka Balasooriya (c.balasooriya@unsw.edu.au)

**Version:** 2  **Date:** 01 Nov 2017

**Author’s response to reviews:**

1st October 2017

Dear Editor

We have attended to the editorial comments noted and described the changes made in relation to each of the comments below. The revised manuscript has been uploaded as a clean copy as requested.

1) Please list the full names, institutional addresses and email addresses for all authors on the title page.

Added.

2) Please rename the "Method" section and "Conclusion" section of the abstract as "Methods" and "Conclusions". Please include a list of three to ten keywords representing the main content of the article after the Abstract.

Amended and added.

3) Please rename the "Conclusion" section as "Conclusions".

Amended.
4) Please move the list of abbreviations after the Conclusions and before the Declarations.

Moved.

5) Please rename the "Consent to publish" subsection as "Consent for publication".

Amended.

6) Please revise your current "Availability of data and materials" statement to reflect one of the formats indicated in our submission guidelines:
https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

Revised.

7) Please include the sentence "All authors have read and approved the final version of this manuscript" in the "Authors' contributions" subsection and ensure that this is the case.

This is the case and we have added the sentence.

8) Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References.

Title removed.

9) At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables and figures should also be clean versions.

Clean versions uploaded.

10) Please rewrite the first three sentences of the "Limitations" section (beginning with "The findings must be considered"); these sections of your manuscript overlap with text in another published paper.

These have been revised.

Yours sincerely,
Dr Reema Harrison